Author's response to reviews

Title: Effect of low frequency magnetic fields on melanoma: tumor inhibition and immune modulation

Authors:

Yunzhong Nie (nie.yunzhong@gmail.com)
Leilei Du (du.lei.good@163.com)
Yongbin Mou (mouyongbin@yahoo.com.cn)
Zhenjun Xu (xjz881225@163.com)
Leihua Weng (weng_2006@yeah.net)
Youwei Du (dyw@nju.edu.cn)
Yanan Zhu (zh-yanan@163.com)
Yayi Hou (yayihou@nju.edu.cn)
Tingting Wang (wangtt@nju.edu.cn)

Version: 5
Date: 26 November 2013

Author's response to reviews:

Dear Editor,

Thanks a lot for helping us publish this manuscript. The questions you mentioned is answered as follows:

1. Drum Tower Hospital is the affiliated hospital of Nanjing University Medical School. All the animal experiments were done in the animal house of Drum Tower Hospital. Therefore, we need the ethics approval from them. It's our fault not make clear the full name of Drum Tower hospital. In the revised manuscript, we have changed it to Nanjing University Medical School affiliated Drum Tower Hospital.

2. We have removed the additional files.

Thanks again for your kind help!

Best!

Tingting Wang